Author's response to reviews

Title: Audio computer-assisted self-interviewing (ACASI) may avert socially desirable responses about infant feeding in the context of HIV

Authors:

Anthony K Waruru (awaruru@ke.cdc.gov)
Ruth Nduati (rnduati@icconnect.co.ke)
Thorkild Tylleskar (thorkild.tylleskar@cih.uib.no)

Version: 8 Date: 26 July 2005

Author's response to reviews: see over
The Editor, Journal of Medical Informatics and Decision Making, Biomed Central,

Dear Editors,

Re: Manuscript for the Journal of Medical Informatics and Decision Making
Please find the reviewed manuscript by Anthony K. Waruru, Ruth Nduati and Thorkild Tylleskär entitled "Audio computer-assisted self-interviewing (ACASI) may avert socially desirable responses about infant feeding in the context of HIV", that we are re-submitting for publication after review.
In response to Dr Elizabeth C. Moylan email requesting for additional formatting, the manuscript is being re-submitted with some editions.

Affiliations – Affiliations were changed according to instructions and author email addresses listed separately.

Keywords section – The keyword section was removed.

Abstract – Unnecessary capitalization was removed.

Page breaks – Page breaks were removed after the title page.
References – Referencing was done using consecutive numbers in square brackets, and superscripting removed. References were placed before the full stop.

Competing interests – There were none to declare and the text was changed to read ‘The authors declare that they have no competing interests’.

References - The reference list includes all authors.

Manuscript sections - Manuscript sections are divided in the following order: Abstract; Background; Methods; Results; Discussion; Conclusions; Competing interests; Authors' contributions; Acknowledgements; References; Figure legends; Tables.

Figure legends – The section now comes immediately after the references section and tables after the figure legends. The section 'figure legends and tables' has been deleted.

Figures - The embedded figure title in figure 1 has been removed.

Figure 1 – The picture of the woman and consent. The figure of the woman has been obscured for anonymity.

Tables - Tables were divided into cells with all borders visible.

Correspondence: Kindly address correspondence on the paper to:
Anthony K. Waruru,
P.O. Box 1578,
040100, Kisumu, KENYA.
Tel: +254 057 530117/8
Cell: +254 0722 841763
email: [awaruru@ke.cdc.gov]